Author's response to reviews

Title: Autonomic neuropathy in Fabry disease: a prospective study using the Autonomic Symptom Profile and cardiovascular autonomic function tests.

Authors:

Marieke Biegstraaten (M.Biegstraaten@amc.uva.nl)
Ivo N. van Schaik (i.n.vanschaik@amc.uva.nl)
Wouter Wieling (w.wieling@amc.uva.nl)
Frits A. Wijburg (f.a.wijburg@amc.uva.nl)
Carla E.M. Hollak (c.e.hollak@amc.uva.nl)

Version: 2 Date: 29 November 2009

Author's response to reviews:

Mrs. M. Norton
Editor-in-chief BMC Neurology
Academic Medical Center
Dept of Neurology, H2-235
PO box 22660
1100 DD Amsterdam, The Netherlands
T: +31 20 5666889 F: +31 20 5669374 E: m.biegstraaten@amc.uva.nl

November, 29, 2009

Dear Mrs. Norton,

In addition to my letter dated 26 November 2009 regarding the submission of our manuscript entitled “Autonomic neuropathy in Fabry disease: a prospective study using the Autonomic Symptom Profile and cardiovascular autonomic function tests”, please find enclosed a figure legend that unfortunately was not included. My apologies for any inconvenience.

We are looking forward to your response to the article.

Yours sincerely,

Marieke Biegstraaten